Author's response to reviews

Title: Clinical prediction of 18F-FDG PET activity in sarcoidosis

Authors:

Rémy LM Mostard (r.mostard@atriummc.nl)
Sander MJ van Kuijk (sander.vankuijk@maastrichtuniversity.nl)
Johny A Verschakelen (johny.verschakelen@uzleuven.be)
Marinus JPG van Kroonenburgh (m.van.kroonenburgh@mumc.nl)
Patty J Nelemans (patty.nelemans@maastrichtuniversity.nl)
Petar AHM Wijnen (petal.wijnen@mumc.nl)
Marjolein Drent (m.drent@hetnet.nl)

Version: 2 Date: 26 April 2012

Author's response to reviews: see over
Dear Editor,

We would like to submit the revised version of the manuscript entitled ‘Clinical prediction of 18F-FDG PET activity in sarcoidosis’ for publication in BMC Pulmonary Medicine.

Enclosed you will find our revised manuscript. We addressed all points raised by the Editorial Office and changed the manuscript accordingly. In the Methods section we added the name of the Ethics Committee and reference number of the approval. After the Abbreviations, we added a Competing interest section, followed by an Authors’ contributions section.

We hope that the revised manuscript fulfils your requirements now and that you will consider this manuscript for publication in BMC Pulmonary Medicine.

Awaiting your reply,
Yours sincerely,

Prof. M. Drent, MD, PhD, professor of interstitial lung diseases
Faculty of Health, Medicine and Life Science, University Maastricht
PO Box 3100, 6202 NC Maastricht, NL
Phone: +31 43 3882087
email: m.drent@maastrichtuniversity.nl